Check for updates

Citation: Zhang Y, Zahreddine M, Abreu K, Dantas MA, Charland K, Pierce L, et al. (2023) Knowledge, attitude and practice (KAP) and risk factors on dengue fever among children in Brazil, Fortaleza: A cross-sectional study. PLoS Negl Trop Dis 17(9): e0011110. https://doi.org/10.1371/journal.pntd.0011110

Editor: Ran Wang, Beijing Children's Hospital Capital Medical University, CHINA

Received: January 27, 2023

Accepted: September 12, 2023

Published: September 25, 2023

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0011110

Copyright: © 2023 Zhang et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Requests for deidentified or anonymized data should be sent to the research ethics board at Université de Montréal, **RESEARCH ARTICLE**

Knowledge, attitude and practice (KAP) and risk factors on dengue fever among children in Brazil, Fortaleza: A cross-sectional study

Yang Zhang¹, Monica Zahreddine², Kellyanne Abreu³, Mayana Azevedo Dantas³, Katia Charland², Laura Pierce², Valéry Ridde⁴, Kate Zinszer^{2,5}*

1 School of Population and Global Health, McGill University, Montreal, Quebec, Canada, 2 Center for Public Health Research, University of Montreal, Montreal, Quebec, Canada, 3 State University of Ceará, Ceará, Brazil, 4 Institute for Research on Sustainable Development, CEPED (IRD-Université de Paris), Université de Paris, ERL INSERM SAGESUD, Paris, France, 5 School of Public Health, University of Montreal, Montreal, Quebec, Canada

* kate.zinszer@umontreal.ca

Abstract

Background

Dengue fever is a mosquito-borne viral disease that is associated with four serotypes of the dengue virus. Children are vulnerable to infection with the dengue virus, particularly those who have been previously infected with a different dengue serotype. Sufficient knowledge, positive attitudes, and proper practices (KAP) are essential for dengue prevention and control. This study aims to estimate the dengue seropositivity for study participants and to examine the association between households' dengue-related knowledge, attitudes, and practices (KAP), and children's risk of dengue seropositivity, while accounting for socioeconomic and demographic differences in Brazil.

Methodology/Principal findings

This analysis was based on a cross-sectional study from Fortaleza, Brazil between November 2019, and February 2020. There were 392 households and 483 participant children who provided a sample of sufficient quality for serological analysis. The main exposure was a household's dengue-related knowledge, attitudes, and practices, assessed through a questionnaire to construct a composite KAP score categorized into three levels: low, moderate, and high. The main outcome is dengue immunoglobulin G(IgG) antibodies, collected using dried blood spots and assessed with Panbio Dengue IgG indirect ELISA (enzyme-linked immunosorbent assays) test commercial kits.

The estimated crude dengue seroprevalence among participating children (n = 483) was 25%. Five percent of households (n = 20) achieved a score over 75% for KAP, sixty-nine percent of households (n = 271) scored between 50% and 75%, and twenty-six percent of households (n = 101) scored lower than 50%. Each KAP domain was significantly and positively associated with the others. The mean percentage scores for the three domains are 74%, 63%, and 39% respectively. We found high household KAP scores were associated

CERSES (email: cerses@umontreal.ca) to ensure that data is shared in accordance with participant consent. Participants were only asked for limited consent to use of the data in related studies conducted by the project investigators and their students, subject to the approval by a Research Ethics Board.

Funding: This research was funded by the Canadian Institutes of Health Research, through the Project Grant program (201803PJT-400444-RC2-CFCA-120159). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

with an increased adjusted relative risk (aRR) of seropositivity (aRR: 2.11, 95% CI: 1.11– 4.01, p = 0.023). Household adult respondents' education level of elementary school or higher was negatively associated with children's risk of being seropositive (aRR: 0.65, 95% CI: 0.48–0.87, p = 0.005). The risk of seropositivity in older children (6–12 years old) was over 6 times that of younger children (2–5 years old) (aRR: 6.08, 95% CI: 3.47–10.64, p<0.001). Children living in households with sealed water tanks or no water storage had a lower risk of being seropositive (aRR: 0.73, 95% CI: 0.54–0.98, p = 0.035).

Conclusions/Significance

Our results provide insight into the prevalence of dengue seropositivity in Fortaleza, Brazil in children, and certain demographic and socioeconomic characteristics associated with children's risk of being seropositive. They also suggest that KAP may not identify those more at-risk for dengue, although understanding and enhancing households' KAP is crucial for effective community dengue control and prevention initiatives.

Author summary

Dengue fever has become an increasing threat to public health, with its global expansion and increased presence in dengue endemic countries. This study provides insight into the prevalence of dengue seropositivity in children in Fortaleza, Brazil, a city which has been a hotspot for arbovirus infections. We estimated seroprevalence for certain study population characteristics and identified characteristics that were associated with an increased risk of dengue seropositivity. We also explored the associations of a composite measure of knowledge, attitudes, and practice (KAP) with seropositivity and found an inverse relationship between KAP for dengue control and children's seropositivity. KAP could change due to previous infection experience, which is a potential limitation of using KAP as a potential predictor of dengue seropositivity in cross-sectional studies. Despite this, KAP remains useful for identifying gaps in knowledge, attitude, and practice that can be used to inform public health measures, such as education campaigns. Measures of social acceptability of interventions should also be considered for inclusion in similar studies, as it would provide an indication of likelihood of adoption of interventions, which provides additional insight for different dengue interventions.

Introduction

Dengue fever is a mosquito-borne viral disease that is associated with four serotypes of the dengue virus [1]. Dengue is an expanding threat to public health globally and the leading cause of serious illness and death among children in several Asian and Latin American countries [1]. Children are vulnerable to infection with the dengue virus, particularly those who have been previously infected with a different dengue serotype [2]. Brazil, the largest country in South America, is experiencing increased dengue incidence, with over 1.5 million cases reported in 2019 [3]. A complex mixture of demographic and socio-economic factors influences the distribution and transmission of dengue [4], which include inadequate living conditions, immuno-logically naïve populations, global trade and population mobility, climate change, and the adaptive nature of the principal mosquito vectors *Aedes aegypti* and *Aedes albopictus* [5–8].

Given that dengue vaccines are not suitable nor available for all dengue-endemic countries and anti-viral medications are not available, control of the mosquito vectors and reducing human-vector contact are recognized as the primary strategies for dengue prevention [9]. *Aedes aegypti* and *Aedes albopictus* are opportunist mosquitoes adapted to urban environments [10]. Precipitation and open water storage create sites for breeding and aquatic stages of the mosquito's life cycle [11]. In several Brazilian cities, residents store clean water in open containers and water reservoirs creating ideal sites for *Aedes* within and near the households [12]. Poor quality housing and sanitation management with high population density are key determinants for the sustained propagation of dengue [13].

Measures of knowledge, attitudes, and practices (KAP) are common and used to identify knowledge gaps, evaluate awareness levels and assess behavioural patterns among members of the community, that may facilitate understanding and public health action [14]. High levels of KAP can empower individuals to participate in disease control and prevention programs [15]. Although there have been several studies assessing KAP and its relationship to dengue risk in different populations, little is known about the link between households' KAP, demographic and socio-economic characteristics and children's dengue seropositivity [16–18]. The objectives of this analysis were to estimate dengue seroprevalence in a pediatric study population and to examine the associations between households' dengue-related KAP and children's dengue seropositivity.

Methods

Ethics statement

Data collection for the study was approved by the University of Montreal's Comité d'éthique de la recherche en sciences et en santé (CERSES) and the Comité de Ética em Pesquisa da Universidade Estadual do Ceará (State University of Ceará, reference number 3.083.892). Approval for the study was obtained from the Ethical Review Committee for Scientific Research of Brazil's National Ethics Council for Research. All parents/guardians gave written informed consent (by signature or fingerprint) for their children to participate.

Study design

This analysis is based on cross-sectional data collected as part of the baseline data collection of a larger international research project [19]. The original project was a pragmatic cluster randomized controlled trial (cRCT) in Fortaleza, Brazil, which aimed to evaluate a community mobilization intervention for dengue control. There are 3,020 census tracts (clusters) in Fortaleza and each cluster was enumerated and filtered by eligibility criteria. Eligible clusters were then categorized according to the dengue vulnerability index (DVI), based on the 2010 Census. To limit contamination among clusters, any selected cluster with common boundaries to previously selected clusters was excluded and replaced by the next cluster (from a ranked list based upon stratification criteria).

The cRCT was powered for the primary objective of comparing seroprevalence between those receiving the intervention and those not receiving the intervention [19]. The current study is part of a secondary objective, therefore a power analysis was not conducted. The cRCT planned to collect data from 68 clusters with a target recruitment of 5,848 children between 2 and 12 years old. For further detail, please see the original trial protocol in Zinszer et. al. 2020 [19]. Due to the Covid-19 pandemic and instability in the study region, the baseline data collection terminated before data from all clusters were gathered. The team was able to obtain the target sample size (n = 86 per cluster) in four clusters although there were eight clusters where this was not possible due to the field conditions during the time period of November 2019 and February 2020. In the incomplete clusters, complete data was collected (e.g., questionnaire and DBS) on a smaller number of participants within these clusters.

Eligibility criteria and study procedure

Households in selected clusters were eligible to participate if there were children between 2 to 12 years old living in the household, and if both the parent or legal guardian gave consent and the child gave assent for the data collection. There were two different informed consent forms (ICF) and one assent form. The first ICF was signed (or digit printed if not able to write) by the person responsible for the household and able to answer the household questionnaire (principal adult respondent). For each participating child, a specific ICF for serology blood collection was signed by the child's parent or guardian. Study objectives and the details of the procedures were explained in adapted language by trained study staff and an assent form was signed or fingerprinted by the participating child if they agreed to participate. Households and children with contaminated blood samples and/or with indeterminate serology results were excluded from this analysis.

Household surveys were administered to the principal adult respondent of each household in Portuguese by a trained interviewer during door-to-door visits. Household surveys included questions about living conditions, socioeconomic status, social capital, and basic health history of the participating children. Respondents were also asked questions related to dengue knowledge, attitudes, and practice (KAP).

Knowledge, attitudes, and practice

The KAP module consisted of partially categorized questions as well as multiple choice, dichotomous and Likert scale responses (S1 Appendix). Five questions in the knowledge domain gauged awareness of dengue symptoms, modes of transmission, and techniques for dengue prevention. Five questions in the attitude domain assessed perceptions about dengue and opinions and acceptability of dengue prevention practices. There were six items in the practice domain related to the adoption of preventive practices and dengue treatment seeking behaviour. A scoring system was applied to create a KAP index. For the multiple choice and dichotomous questions, only correct responses were assigned positive scores (an incorrect response was given a score of zero). For Likert scale questions, the highest score was assigned for 'strongly agreed', 0 for 'strongly disagree' with gradations of 0.5 for 5-choice Likert scale questions. The maximum scores of 10, 8, and 10.5 points were applied to the knowledge, attitude, and practice domain, respectively. Households' total KAP score (maximum of 28.5 points) were categorized into groups: low (<15 points or <50% of maximum points), moderate (>15 or <22 points or between 50% to 75% of maximum score) or high (>22 points or >75% of the maximum score) [17,20,21].

Seroprevalence test

Dried blood spot samples were collected from children by a finger prick performed by a trained nurse technician. All samples were placed on filter paper and were anonymized with a study ID code during collection. Three circles of 15 mm each were filled with capillary blood and allowed to dry at least 2 to 6 hours at room temperature, avoiding exposure to sunlight. Once dried, each card was packaged separately in a hermetic plastic bag containing silica desiccant. DBS samples were then refrigerated until assessment for quality control purposes, standards of sufficient quantity of blood or layering of blood drops is available elsewhere [22], and then stored in a freezer at -20°C. At the laboratory, three punches of 6 mm from the DBS were soaked in a 1.5 mL vial containing 250 µL of eluent solution, shaken overnight at 4°C to ensure

adequate serum elution. The next day, the eluates were centrifuged at 2,500 rpm for 2 minutes in a refrigerated microfuge and then transferred to a clean and sterile tube. Undiluted DBS eluates were treated as a 1:25 equivalent dilution of human serum, based on previous work that demonstrated that approximately three 6 mm circles contained 21 µL of dried blood, corresponding to approximately 10.5 µL of serum that were eluted in 250 µL of the eluent solution. Dengue seropositivity was defined as the detection of immunoglobulin G (IgG) antibodies to dengue antigen serotypes (1, 2, 3, and 4) using the Panbio Dengue IgG indirect ELISA (enzyme-linked immunosorbent assays), a commercial capture ELISA with high specificity and sensitivity, and excellent correlation with Hemagglutination Inhibition (HAI) assay [23,24]. To ensure high-quality data collection, a field supervisor was present for both blood collection and questionnaire administration.

Statistical analyses

The primary outcome of this study was dengue seropositivity (yes/no). Seroprevalence with 95% confidence intervals were estimated using a single-proportion z-test. We conducted Chisquared tests to examine the relationship of each domain of KAP (low, moderate, high) with dengue seropositivity. Kendall rank correlation tests were used to assess the associations between each pair of KAP domains (i.e., K and A, K and P, A and P). The associations between sociodemographic covariates and seropositivity were assessed through univariate regression. A multivariable regression model was then developed to assess associations between risk of seropositivity and potential predictors of risk including KAP, rainy season, age, race, average monthly household income, household water storage method, household's recent dengue episode and dengue death history, and the education level and occupation status of the household's principal adult respondent. The selection of variables was guided by hypothesized causal associations and confounders through literature review and direct acyclic graphs. To address any overdispersion by clustering we considered a random effects model and a quasibinomial model but the appropriateness of the random effects model was limited by the small cluster sizes and small number of clusters and the quasibinomial model dispersion parameter was 1.0 [25], indicating little evidence of overdispersion. For both univariable and multivariable regression models, relative risks (RR) and 95% confidence intervals were estimated based on Zou's modified Poisson regression approach [26]; an alternative to logistic regression that directly estimates relative risk, rather than odds ratios, and is appropriate when the outcome is common [27].

Children's ages were divided into two groups: 2–5 years old and 6 to 12 years old. Children's race was classified into five categories, based upon the classification used by the *Instituto Brasileiro de Geografia e Estatística* which is used for the Brazilian census: *Branco* (White), *Pardo ou moreno* (Multiracial), *Preto* (Black), *Asiático* (Asian), and *Indígena* (Indigenous) [28]. Race was reduced to two categories, multiracial and white vs. other, for the multivariable regression to ensure sufficient representation in the race categories.

To consider the climate factor, we separated the children according to the date their household was surveyed and the date the blood sample was collected (both occurring on the same day), using cumulative monthly rainfall [29]. We distinguished between households with no potable water storage and sealed individual water tanks from households with open water storage methods (e.g., buckets, pots) [30]. Households were considered to have a dengue history if they experienced a recent dengue episode or the death of a family member due to dengue. Household education level of the adult respondent was categorized based upon completion of elementary school as elementary school is essential for an individual's literacy. There were two categories of occupation status, respondents who had paid employment at the time of the questionnaire vs being unemployed, retired, or other (e.g., refused to answer). Households' monthly income were categorized as low (<500 Brazilian real), moderate (>500 or <2500 Brazilian real), or high (>2500 Brazilian real) based on Fortaleza's cost of living and the average household income of the resident population from the Ceara state [28,31]. MICE imputation was carried out for the multivariable regression analysis to impute missing household income and children's race.

All statistical analyses were conducted using R version 4.2.0.

Results

Characteristics of participating households and children

There were 392 households and 483 children who provided a blood sample of sufficient quality for this analysis. The mean age of participating children was 6.5 years with an almost equal proportion of boys and girls participating in the study (Table 1). The mean age (SD) of household principal adult respondents was 29 years with the majority being female (n = 345, 88%). Children were most often reported as being multiracial/white (n = 434, 90%). Parental respondents had a high completion rate of elementary school (n = 276, 70%) and most households had a moderate household income (n = 283, 72%). Household and children's sample size per borough, and characteristics of the study population after imputation can be found in <u>S2</u> and <u>S3 Appendices</u> respectively.

Table 2 shows crude seroprevalence, seroprevalence differences and ratios for children by individual characteristics. There were 121 seropositive samples and 362 seronegative samples, resulting in an estimated crude seroprevalence of 25.1% (95% CI: 21.3% - 29.2%).

Knowledge, attitudes, practices and overall KAP

Only five percent (n = 20) of households achieved over 75% for overall KAP, while most households achieved scores between 50% and 75% (n = 271, 69%), and 101 households scored lower than 50% of the total score (26%). Table 3 shows the result of the Kendall rank correlation tests between each KAP domain and the result of Chi-squared tests with children's seropositivity. The results of the Kendall rank correlation tests and Chi-squared tests demonstrate that the three domains have a statistically significant positive association with each other. However, in univariate tests, there was no evidence of an association between children's seropositivity and the household's level of each individual domain (knowledge, practice, or attitude) or with the overall KAP measure.

For each KAP question, the percentage score was calculated by dividing the score of each household achieved by the maximum points of each question (S4 Appendix). Out of the three domains, the knowledge domain had the highest mean percentage score of 74%, followed by the attitude domain and then the practice domain at 63% and 39%, respectively. While all questions in the knowledge domain have a mean percentage score over 65%, the mean percentage score varies largely among questions in the attitude and practice domains.

Demographic and socioeconomic determinants of dengue seropositivity in children

In the multivariable regression model, the parental education level, households' KAP level, age of the child, and household water storage method were shown to be significantly associated with the risk of dengue seropositivity in children (Table 4).

Among all included predictors, children's age was most strongly associated with the risk of dengue seropositivity, which was significantly higher among children aged 6–12 years

Children	Total N (%)
Child Age, year	
Mean (SD)	6.5 (2.7)
Median [Min, Max]	6.0 [2.0, 12.0]
2-5	181 (50%)
6-12	181 (50%)
Child Race	
Multiracial/White	434 (89.9%)
Other	49 (10.1%)
Child Gender	
Female	222 (46.0%)
Male	261 (54.0%)
School Attendance	
Yes	431 (89.2%)
No	50 (10.4%)
Refused	2 (0.4%)
Households	Total (N=392)
Household Monthly Income (BRL)*	
Low	21 (5.4%)
Moderate	254 (64.8%)
High	79 (20.2%)
Nonresponse	38 (9.7%)
Respondent Education	
Lower than Elementary School	116 (29.6%)
Elementary School or Higher	276 (70.4%)
Respondent Gender	
Female	348 (88.8%)
Male	44 (11.2%)
Does not know	3 (0.8%)
Respondent Race	
Black	28 (9.7%)
Multiracial	285 (72.7%)
White	55 (14.0%)
Asian	10 (2.6%)
Other	4 (1.0%)
Occupation	
Retired/Unemployed/Other	202 (51.5%)
Employed	190 (48.5%)
Household KAP	
Low	101 (25.8%)
Moderate	271 (69.1%)
High	20 (5.1%)

Table 1. Sociodemographic Characteristics of Children and Households, total children participants N = 483.

*BRL = Brazilian real

https://doi.org/10.1371/journal.pntd.0011110.t001

compared to children aged 2–5 years old (RR: 6.1, 95% CI: 3.5–10.7, p<0.001). Children of parents with higher education levels, who at least completed elementary school, were less likely to be seropositive (RR: 0.7, 95% CI: 0.5–0.9, p = 0.005). There were no statistically significant

	Total No.	Seroprevalence, %	95% CI	Crude prevalence ratio (95% CI)
Total	483	25.1	21.3-29.2	NA
Child Age, year				
2–5	193	6.2	3.4-10.9	1 [Reference]
6-12	290	37.6	32.0-43.5	6.1 (3.4–10.7)
Child Race*				
Multiracial/White	440	25.0	21.0-29.3	1 [Reference]
Other	43	25.6	13.5-41.2	1.0 (0.5–2.1)
Child Gender				
Female	222	27.0	21.4-33.5	1 [Reference]
Male	261	23.3	18.5-29.1	0.8 (0.5–1.2)
School Attendance				
Yes	431	27.4	23.3-31.9	1 [Reference]
No	50	4.0	0.7-14.9	0.1 (0.0–0.5)
Refused	2	50.0	9.5-90.5	2.7 (0.2-42.8)

Table 2. Crude Seroprevalence and Seroprevalence Differences and Ratios for Children by Individual Characteristics.

* "Does not know" responses for Child Race were imputed

https://doi.org/10.1371/journal.pntd.0011110.t002

associations detected between dengue seropositivity and monthly household income, principal adult's occupation, children's race, household's dengue history or rainy season. Children living in households without potable water storage or using sealed water tanks were found to be significantly associated with a lower risk of dengue seropositivity (RR: 0.7, 95% CI: 0.5–1.0, p = 0.038). Notably, our result showed that higher values of KAP were associated with an increased risk of dengue seropositivity (RR: 2.1, 95% CI: 1.1–4.0, p = 0.024).

Discussion

The results of our study demonstrated that one quarter of the participating children were seropositive for dengue and the results identified certain socioeconomic and demographic characteristics that were associated with the risk of dengue seropositivity. The results also suggest that KAP was not a protective predictor of dengue seropositivity risk in our cross-sectional study, as it was positively associated with seropositivity.

Table 3. Kendall Rank Correlation and Chi-Square Test Summary.

	tau ^[1]	I	o-value
Attitude vs. Practice	0.12		0.008
Practice vs. Knowledge	0.13	0.003	
Knowledge vs. Attitude	0.13	0.002	
	Chi-square	df ^[2]	p-value
Attitude vs. Serology Result	2.32	2	0.313
Practice vs. Serology Result	5.43	2	0.066
Knowledge vs. Serology Result	1.63	2	0.442
Overall KAP vs. Serology Result	3.75	2	0.153

^[1]₋tau = Kendall's tau statistic

^[1]₋tau = Kendall's tau statistic

^[1]tau = Kendall's tau statistic

 $^{[2]}_{-}$ df = Degree of freedom

https://doi.org/10.1371/journal.pntd.0011110.t003

Characteristic	Levels	RR ¹	95%CI ¹	p-value
Household KAP	Low	REF	REF	REF
	Moderate	1.23	[0.86, 1.75]	0.267
	High	2.11	[1.11, 4.02]	0.024*
Respondent's Education	Less than Elementary School	REF	REF	REF
	Elementary School or Higher	0.65	[0.48, 0.87]	0.004**
Child Age	Younger Group (2–5 years old)	REF	REF	REF
	Older Group (6-12 years old)	6.09	[3.47, 10.69]	< 0.001***
Child Race	Other	REF	REF	REF
	Multiracial/White	1.03	[0.63, 1.69]	0.915
Water Storage Method	Other Water Storage Method	REF	REF	REF
	Sealed Water Tank/No Water Storage	0.73	[0.55, 0.98]	0.038*
Respondent's Occupation	Retired/Unemployed/Other	REF	REF	REF
	Employed	0.75	[0.56, 1.02]	0.065
Household Dengue Experience	Yes	REF	REF	REF
	No	0.80	[0.58, 1.10]	0.18
Household's Monthly Income	Low	REF	REF	REF
	Moderate	0.96	[0.56, 1.65]	0.89
	High	1.26	[0.68, 2.30]	0.460
Rainy Season	No	REF	REF	REF
	Yes	1.11	[0.84, 1.48]	0.462

Table 4. Multivariable Regressio	n Model of Sociodemographic Predic	tors of Children's Serology Results.

¹ RR = Relative Risk, CI = Confidence Interval

https://doi.org/10.1371/journal.pntd.0011110.t004

The overall dengue seroprevalence was comparable to other dengue seroprevalence studies of children in dengue-endemic settings [32,33]. We also found that the seroprevalence varied widely between different sociodemographic factors such as race, sex, and age. Black and white children had the highest seroprevalence compared to multiracial and other groups although given the low number of seropositive children in certain racial groups, these differences should be interpreted with caution. Associations between racial and ethnic minority status and dengue seropositivity have been found by other studies [16,34,35], and race is often related to increased vulnerability of disease due to social conditions including housing, municipal services, and education [36].

Girls were estimated to have a higher dengue seroprevalence compared to boys, although the difference was not statistically significant. There are a mixture of findings in terms of gender differences in dengue incidence, some have found greater male dengue incidence among both children and adults which is likely due to gender-related differences in exposure such as time away from home and participation in outdoor games [36–39], while others have found no differences [40,41]. In our study, we also found that children's age was positively associated with seropositivity, which is comparable to many similar studies [32,42,43]. Considering IgG remains detectable in blood for many years [44], children in the older group (6–12 years old) had a longer time at risk of infection compared to younger children. Previous studies have determined that the majority of dengue virus transmission occurs within communities or schools, school attendance could play an important role in the progression of dengue epidemics given the increases in the dengue infection rate following the beginning of the school year [38,45,46].

Children were 35% less likely to be seropositive if the parent/guardian respondent had at least completed elementary school, compared to those children whose parent/guardian had

lower educational attainment. Multiple studies have demonstrated that education was positively associated with the use and promotion of dengue prevention, such as ability to take actions of control, acceptability of prevention measures and educating family members and neighbors [47,48]. Our study also suggests that water storage methods are important in terms of risk of dengue seropositivity in children, which has been found by several other studies [49– 51]. Open containers and unsealed reservoirs can act as important breeding sites for *Aedes* mosquitos [12,49,51], while piped water and sealed reservoirs can reduce the presence of *Aedes* mosquitos significantly [12,49].

An interesting finding from our study is that children living in households with high KAP levels had a higher relative risk of dengue seropositivity. One hypothesis for this finding is related to our outcome measure and an important limitation of cross-sectional data—a lack of temporality. We did not measure active dengue infections but measured dengue seropositivity using IgG, which can persist in the blood for several years [44]. It is plausible that this inverse association is due to a mismatch of temporality between our outcome and exposure of interest.

Cohort and case-control studies have found participants who had experience with dengue had a significantly higher knowledge score, as they are likely to seek for information when encountering the disease or have more chances to learn from healthcare staff when receiving treatments [52–54]. Among children with previous dengue, parents/guardians have been more likely to perceive dengue as a serious concern [46]. Studies conducted in Indonesia and Colombia have shown that adults who have experienced dengue are also more likely to have a positive attitude toward preventing it [55,56]. In terms of the impact of dengue experience on prevention practices, a study from Malaysia found age and dengue history are the main determinants that influence a high practice level [57], which is consistent with another study's result where previous dengue infections were positively associated with dengue prevention practices [58]. Therefore, if a child had a previous dengue infection, it is likely that this experience improved the overall KAP in the household [52,58].

Participating households demonstrated a good understanding of vector control and dengue transmission, prevention, and symptoms, as evidenced by the relatively high overall mean percentage score in the knowledge domain compared to the knowledge level observed in previous studies conducted in dengue pandemic areas [20,59,60]. This phenomenon can likely be attributed to the high prevalence of dengue fever in Brazil and exposure to dengue prevention education campaigns and related information.

In terms of attitudes, a fair overall mean score was observed. Almost all participating respondents believe they were at-risk of dengue and that dengue can be prevented although the majority of participants lacked confidence in the community's dengue prevention capabilities. This attitude could be due to several factors, which include a lack of support from government and community services [61–63], and disillusionment about household and community agency [64,65]. There was poor level of household-level vector control practices and in certain studies, it was demonstrated that knowledge was not associated with any significant behavior change [66,67]. Given the low confidence in community prevention ability among the study population, community engagement activities may be necessary to deliver key messages on dengue and to work with the community to identify acceptable vector control interventions [66,68–70]. This would ensure the sustainability of long-term community dengue prevention interventions.

Limitations

The study has several limitations. The main limitation is the cross-sectional design with exposure and outcome measured at the same time, making it impossible to establish temporality in the associations. KAP can change substantially over time and in response to certain events, such as previous dengue infection. There is likely selection bias, as the response rate varied by clusters and decreased throughout the study due to the political context. Information bias could have occurred with the questionnaire data collection given that it was based on self-reported information, particularly for the KAP questions, although we assume that the impact would be non-differential. While the diagnostic test Panbio Dengue IgG Indirect ELISA is proven to be highly reliable and showed high agreement with HAI [23,24], there is a possibility that some participants were misclassified in terms of seropositivity (false negative) with 95.2%-97.9% sensitivity for secondary infections (77.1% sensitivity for primary) and 93.4%-100% specificity [22,23,71]. This may have led to misclassification of serostatus. Confounding bias could have occurred via residual confounding as we reduced the number of categories within certain variables to improve the precision of our results. In terms of race and reported household income, although may not have been representative for study characteristics [28,31].

Conclusion

We found certain demographic and socio-economic characteristics, particularly children's age, parent's education, and water storage method were associated with children's risk of sero-positivity. Further research is needed to identify barriers that influence attitudes and practice among the study population and to understand if and how community mobilization is an effective approach for dengue control. We also identified an important limitation of using KAP measures for cross-sectional studies for dengue infection: KAP could change due to previous infection experience, which is likely relevant for other infectious diseases. Despite this limitation of using KAP as a potential predictor of dengue seropositivity in cross-sectional studies, it remains useful for identifying gaps in knowledge, attitude, and practice that can be used to inform public health measures, such as education campaigns. Measures of social acceptability of interventions should also be considered for inclusion in similar studies [72], as it would provide an indication of likelihood of adoption of interventions, which provides additional insight for different dengue interventions.

Supporting information

S1 Appendix. KAP Survey Questionnaire and Scoring. (DOCX)

S2 Appendix. Household and Children's Sample Size per Borough. (DOCX)

S3 Appendix. Sociodemographic Characteristics of Children and Households After Imputation.

(DOCX)

S4 Appendix. Mean Percentage Score of KAP Questions with 95% Confidence Interval. (DOCX)

Author Contributions

Conceptualization: Laura Pierce, Valéry Ridde, Kate Zinszer.

Data curation: Yang Zhang, Monica Zahreddine, Kellyanne Abreu, Mayana Azevedo Dantas.

Formal analysis: Yang Zhang, Katia Charland, Laura Pierce.

Funding acquisition: Valéry Ridde, Kate Zinszer.

Investigation: Monica Zahreddine, Kellyanne Abreu, Mayana Azevedo Dantas.

Methodology: Yang Zhang, Katia Charland, Kate Zinszer.

Project administration: Monica Zahreddine, Kellyanne Abreu, Mayana Azevedo Dantas, Kate Zinszer.

Supervision: Monica Zahreddine, Kellyanne Abreu, Mayana Azevedo Dantas, Kate Zinszer.

Validation: Yang Zhang, Monica Zahreddine.

Visualization: Yang Zhang.

Writing - original draft: Yang Zhang, Kate Zinszer.

Writing – review & editing: Yang Zhang, Monica Zahreddine, Katia Charland, Laura Pierce, Valéry Ridde, Kate Zinszer.

References

- World Health Organization(WHO). Dengue and Severe Dengue [Internet]. World Health Organization; 2018a. [cited 2022 Oct 24]. Available from: https://www.who.int/en/news-room/fact-sheets/detail/ dengue-and-severe-dengue.
- Jain A, Chaturvedi UC. Dengue in infants: An overview. FEMS Immunology & Medical Microbiology. 2010; 59(2):119–30. https://doi.org/10.1111/j.1574-695X.2010.00670.x PMID: 20402771
- Junior JB, Massad E, Lobao-Neto A, Kastner R, Oliver L, Gallagher E. Epidemiology and costs of Dengue in Brazil: A systematic literature review. International Journal of Infectious Diseases. 2022; 122:521–8. https://doi.org/10.1016/j.ijid.2022.06.050 PMID: 35793756
- Watts MJ, Kotsila P, Mortyn PG, Sarto i Monteys V, Urzi Brancati C. Influence of socio-economic, demographic and climate factors on the regional distribution of Dengue in the United States and Mexico. International Journal of Health Geographics. 2020; 19(1). <u>https://doi.org/10.1186/s12942-020-00241-1</u> PMID: 33138827
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013; 496(7446):504–7. Epub 20130407. https://doi.org/10.1038/ nature12060 PMID: 23563266; PubMed Central PMCID: PMC3651993.
- 6. Morin CW, Comrie AC, Ernst K. Climate and dengue transmission: evidence and implications. Environ Health Persp. 2013; 121(11–12): 1264. https://doi.org/10.1289/ehp.1306556 PMID: 24058050
- Staples JE, Fischer M. Chikungunya virus in the Americas—what a vectorborne pathogen can do. New Eng J of Med. 2014; 371(10): 887–889. https://doi.org/10.1056/NEJMp1407698 PMID: 25184860
- Kraemer MU, Sinka ME, Duda KA, Mylne AQ, Shearer FM, Barker CM, et al. The global distribution of the arbovirus vectors Aedes aegypti and Ae. albopictus. Elife. 2015; 4:e08347. Epub 20150630. https:// doi.org/10.7554/eLife.08347 PMID: 26126267; PubMed Central PMCID: PMC4493616.
- Ritchie SA, Devine GJ, Vazquez-Prokopec GM, Lenhart AE, Manrique-Saide P, Scott TW. 4. insecticide-based approaches for dengue vector control. Ecology and Control of Vector-borne Diseases. 2021;59–89.
- Villamil-Gómez WE, González-Camargo O, Rodriguez-Ayubi J, Zapata-Serpa D, Rodriguez-Morales AJ. Dengue, chikungunya and Zika co-infection in a patient from Colombia. J Infect Public Health. 2016; 9(5):684–6. https://doi.org/10.1016/j.jiph.2015.12.002 PMID: 26754201
- Nakkhara P, Chongsuvivatwong V, Thammapalo S. Risk factors for symptomatic and asymptomatic chikungunya infection. Trans R Soc Trop Med Hyg. 2013; 107(12):789–96. https://doi.org/10.1093/ trstmh/trt083 PMID: 24052594
- Garcia Sánchez DC, Pinilla GA, Quintero J. Ecological characterization of Aedes aegypti larval habitats (Diptera: Culicidae) in artificial water containers in Girardot, Colombia. J Vector Ecol. 2017; 42(2): 289– 97. https://doi.org/10.1111/jvec.12269 PMID: 29125250
- Zinszer K, Degroote S, Ridde V. Interventions for vector borne diseases focused on housing and hygiene in urban areas: a scoping review. Infect Dis Poverty. 2018; 7:96. <u>https://doi.org/10.1186/s40249-018-0477-5</u> PMID: 30173670

- DaBreo S. Inniss-Springer E. Knowledge, Attitudes and Practices study: Baseline for the SMART Health Care Facilities in the Eastern Caribbean Project–Phase II [Internet]. Pan American Health Organization; 2016 [cited 2022 October 24] Available from: https://www3.paho.org/disasters/dmdocuments/ RevisedBaselineKAPReport-SMARTHealthCareFacilities-Phase%20II-11%20NOV2016.pdf.
- Wang J, Chen L, Yu M, He J. Impact of knowledge, attitude, and practice (kap)-based rehabilitation education on the KAP of patients with intervertebral disc herniation. Annals of Palliative Medicine. 2020; 9 (2):388–93. https://doi.org/10.21037/apm.2020.03.01 PMID: 32233633
- Selvarajoo S, Liew JW, Tan W, Lim XY, Refai WF, Zaki RA, et al. Knowledge, attitude and practice on Dengue Prevention and Dengue Seroprevalence in a dengue hotspot in Malaysia: A cross-sectional study. Scientific Reports. 2020; 10(1).
- Guad RM, Sim MS, Wu YS, Aung YN, Low WY, Sekaran SD, et al. Knowledge, attitude, and prevention practice on Dengue and dengue seroprevalence in a dengue hotspot in Sabah, East Malaysia. Research Square[Preprint]. 2021[cited 2023 May 5]. Available from: https://www.researchsquare.com/ article/rs-1066975/v1 https://doi.org/10.21203/rs.3.rs-1066975/v1
- Soghaier MA, Himatt S, Osman KE, Okoued SI, Seidahmed OE, Beatty ME, et al. Cross-sectional community-based study of the socio-demographic factors associated with the prevalence of dengue in the eastern part of Sudan in 2011. BMC Public Health. 2015; 15(1). https://doi.org/10.1186/s12889-015-1913-0 PMID: 26084275
- Zinszer K, Caprara A, Lima A, Degroote S, Zahreddine M, Abreu K, et al. Sustainable, healthy cities: Protocol of a mixed methods evaluation of a cluster randomized controlled trial for Aedes control in Brazil using a community mobilization approach. Trials. 2020; 21(1). https://doi.org/10.1186/s13063-019-3714-8 PMID: 32059693
- 20. Jayawickreme KP, Jayaweera DK, Weerasinghe S, Warapitiya D, Subasinghe S. A study on knowledge, attitudes and practices regarding dengue fever, its prevention and management among dengue patients presenting to a tertiary care hospital in Sri Lanka. BMC Infectious Diseases. 2021; 21(1).
- Ajibola L-SA1, Shohaimi S, Adam M, Nadzir M, Segun O. Systematic review of knowledge, attitude, and practices regarding dengue in Malaysia. Journal of Applied Pharmaceutical Science. 2018; 8 (12):80–91. https://doi.org/10.7324/japs.2018.81221
- 22. World Health Organization. Manual Blood Collection and Handling–Dried Blood Spot (DBS) [Internet]. World Health Organization; 2005 [cited 2020 Feb 20] Available from: https://www.who.int/diagnostics_laboratory/documents/guidance/pm_module14.pdf.
- BioScience. DENGUE IgG INDIRECT ELISA Cat No. E-DEN01G [Internet]. 2023 [cited 2023 Aug 22]. Available from: http://www.rootbio.com/admin/download/2009216131816609.pdf.
- Lopez AL, Adams C, Ylade M, Jadi R, Daag JV, Molloy CT, et al. Determining dengue virus serostatus by indirect IGG Elisa compared with focus reduction neutralisation test in children in Cebu, Philippines: A prospective population-based study. The Lancet Global Health. 2021; 9(1). <u>https://doi.org/10.1016/ S2214-109X(20)30392-2 PMID: 33212030</u>
- Moineddin R, Matheson FI, Glazier RH. A simulation study of sample size for multilevel logistic regression models. BMC Medical Research Methodology. 2007; 7(1). <u>https://doi.org/10.1186/1471-2288-7-34</u> PMID: 17634107
- Zou G. A modified Poisson regression approach to prospective studies with Binary Data. American Journal of Epidemiology. 2004; 159(7):702–6. https://doi.org/10.1093/aje/kwh090 PMID: 15033648
- Gallis JA, Turner EL. Relative measures of association for binary outcomes: Challenges and recommendations for the Global Health Researcher. Annals of Global Health. 2019; 85(1). https://doi.org/10.5334/aogh.2581 PMID: <u>31807416</u>
- Instituto Brasileiro de Geografia e Estatística (Brazilian Institute for Geography and Statistics). 2010 Census [Internet]. [cited 2023 May 8]. Available from: https://www.ibge.gov.br/en/statistics/social/labor/ 18391-2010-population-census.html.
- 29. Spark Weather. Climate and Average Weather Year Round in Fortaleza Brazil [Internet]. [cited 2023 May 8]. Available from: https://weatherspark.com/y/31123/Average-Weather-in-Fortaleza-Brazil-Year-Round.
- Pinchoff J, Silva M, Spielman K, Hutchinson P. Use of effective lids reduces presence of mosquito larvae in household water storage containers in urban and peri-urban zika risk areas of Guatemala, Honduras, and El Salvador. Parasites & Vectors. 2021; 14(1). <u>https://doi.org/10.1186/s13071-021-04668-8</u> PMID: 33741050
- Numbeo. Cost of living in Fortaleza [Internet]. 2022 [cited 2022 Dec 8]. Available from: https://www. numbeo.com/cost-of-living/in/Fortaleza.
- Shah PS, Alagarasu K, Karad S, Deoshatwar A, Jadhav SM, Raut T, et al. Seroprevalence and incidence of primary dengue infections among children in a rural region of Maharashtra, Western India. BMC Infectious Diseases. 2019; 19(1).

- 33. Chiaravalloti-Neto F, da Silva RA, Zini N, da Silva GC, da Silva NS, Parra MC, et al. Seroprevalence for dengue virus in a hyperendemic area and associated socioeconomic and demographic factors using a cross-sectional design and a geostatistical approach, state of São Paulo, Brazil. BMC Infectious Diseases. 2019; 19(1).
- Bde Sierra, Kourí G Guzmán MG. Race: A risk factor for dengue hemorrhagic fever. Archives of Virology. 2006; 152(3):533–42. https://doi.org/10.1007/s00705-006-0869-x PMID: 17106622
- 35. Paradies Y, Ben J, Denson N, Elias A, Priest N, Pieterse A, et al. Racism as a determinant of health: A systematic review and meta-analysis. PLOS ONE. 2015; 10(9). https://doi.org/10.1371/journal.pone. 0138511 PMID: 26398658
- 36. Piedrahita LD, Agudelo Salas IY, Marin K, Trujillo AI, Osorio JE, Arboleda-Sanchez SO, et al. Risk factors associated with dengue transmission and spatial distribution of high seroprevalence in schoolchildren from the urban area of Medellin, Colombia. Canadian Journal of Infectious Diseases and Medical Microbiology. 2018; 2018:1–11. https://doi.org/10.1155/2018/2308095 PMID: 30245759
- Anker M, Arima Y. Male-female differences in the number of reported incident dengue fever cases in six Asian countries. Western Pacific Surveillance and Response. 2011; 2(2). <u>https://doi.org/10.5365/ WPSAR.2011.2.1.002</u> PMID: 23908884
- Ooi E-E, Goh K-T, Gubler DJ. Dengue prevention and 35 years of Vector Control in Singapore. Emerging Infectious Diseases. 2006; 12(6):887–93. <u>https://doi.org/10.3201/10.3201/eid1206.051210</u> PMID: 16707042
- Verma RK, Kumar M, Nirjhar S, Singh M. Dengue in children and young adults, a cross-sectional study from the western part of Uttar Pradesh. Journal of Family Medicine and Primary Care. 2020; 9(1):293. https://doi.org/10.4103/jfmpc.jfmpc_770_19 PMID: 32110607
- 40. Sabchareon A, Sirivichayakul C, Limkittikul K, Chanthavanich P, Suvannadabba S, Jiwariyavej V, et al. Dengue infection in children in Ratchaburi, Thailand: A cohort study. I. Epidemiology of symptomatic acute dengue infection in children, 2006–2009. PLoS Neglected Tropical Diseases. 2012; 6(7). https:// doi.org/10.1371/journal.pntd.0001732 PMID: 22860141
- Astuti EP, Dhewantara PW, Prasetyowati H, Ipa M, Herawati C, Hendrayana K. Paediatric dengue infection in Cirebon, Indonesia: A temporal and spatial analysis of notified dengue incidence to inform surveillance. Parasites & Vectors. 2019; 12(1). <u>https://doi.org/10.1186/s13071-019-3446-3</u> PMID: 31036062
- Thai KT, Nishiura H, Hoang PL, Tran NT, Phan GT, Le HQ, et al. Age-specificity of clinical dengue during primary and secondary infections. PLoS Neglected Tropical Diseases. 2011; 5(6). <u>https://doi.org/10.1371/journal.pntd.0001180</u> PMID: 21713018
- 43. Ng RJ, Chong ZL, Abdul Mutalip MH, Ng C-W. Dengue seroprevalence and factors associated with dengue seropositivity in Petaling District, Malaysia. International Journal of Environmental Research and Public Health. 2022; 19(12):7170. https://doi.org/10.3390/ijerph19127170 PMID: 35742419
- 44. Rossen RD, Birdsall HH. Allergy and immunology. Medical Secrets. 2012;291–343.
- Hernández-Suárez CM, Mendoza-Cano O. Empirical evidence of the effect of school gathering on the dynamics of dengue epidemics. Global Health Action. 2016; 9(1):28026. https://doi.org/10.3402/gha. v9.28026 PMID: 26743450
- **46.** Endy TP. Spatial and temporal circulation of dengue virus serotypes: A prospective study of Primary School Children in Kamphaeng Phet, Thailand. American Journal of Epidemiology. 2002; 156(1):52–9. https://doi.org/10.1093/aje/kwf006 PMID: 12076888
- Diaz-Quijano FA, Martínez-Vega RA, Rodriguez-Morales AJ, Rojas-Calero RA, Luna-González ML, Díaz-Quijano RG. Association between the level of education and knowledge, attitudes and practices regarding dengue in the Caribbean region of Colombia. BMC Public Health. 2018; 18(1). <u>https://doi.org/ 10.1186/s12889-018-5055-z PMID: 29338712</u>
- Murray N, Jansarikij S, Olanratmanee P, Maskhao P, Souares A, Wilder-Smith A, et al. Acceptability of impregnated school uniforms for dengue control in Thailand: A mixed methods approach. Global Health Action. 2014; 7(1):24887. https://doi.org/10.3402/gha.v7.24887 PMID: 25183313
- 49. Telle O, Nikolay B, Kumar V, Benkimoun S, Pal R, Nagpal B, et al. Social and environmental risk factors for dengue in Delhi City: A retrospective study. PLOS Neglected Tropical Diseases. 2021; 15(2). <u>https:// doi.org/10.1371/journal.pntd.0009024</u> PMID: 33571202
- Soghaier MA, Mahmood SF, Pasha O, Azam SI, Karsani MM, Elmangory MM, et al. Factors associated with dengue fever IGG sero-prevalence in South Kordofan State, Sudan, in 2012: Reporting prevalence ratios. Journal of Infection and Public Health. 2014; 7(1):54–61. https://doi.org/10.1016/j.jiph.2013.07. 008 PMID: 24210245
- 51. Vannavong N, Seidu R, Stenström T-A, Dada N, Overgaard HJ. Effects of socio-demographic characteristics and household water management on Aedes aegypti production in suburban and rural villages

in Laos and Thailand. Parasites & Vectors. 2017; 10(1). https://doi.org/10.1186/s13071-017-2107-7 PMID: 28376893

- 52. Suwanbamrung C, Saengsuwan B, Sangmanee T, Thrikaew N, Srimoung P, Maneerattanasak S. Knowledge, attitudes, and practices towards dengue prevention among primary school children with and without experience of previous dengue infection in Southern Thailand. One Health. 2021; 13:100275. https://doi.org/10.1016/j.onehlt.2021.100275 PMID: 34159247
- Wong LP, Shakir SM, Atefi N, AbuBakar S. Factors affecting dengue prevention practices: Nationwide survey of the malaysian public. PLOS ONE. 2015; 10(4). https://doi.org/10.1371/journal.pone.0122890 PMID: 25836366
- Pérez-Guerra CL, Zielinski-Gutierrez E, Vargas-Torres D, Clark GG. Community beliefs and practices about dengue in Puerto Rico. Revista Panamericana de Salud Pública. 2009; 25(3):218–26. https://doi. org/10.1590/s1020-49892009000300005 PMID: 19454149
- 55. Harapan H, Rajamoorthy Y, Anwar S, Bustamam A, Radiansyah A, Angraini P, et al. Knowledge, attitude, and practice regarding dengue virus infection among inhabitants of Aceh, Indonesia: A cross-sectional study. BMC Infectious Diseases. 2018;18(1). https://doi.org/10.1186/s12879-018-3006-z PMID: 29486714
- Benítez-Díaz L, Diaz-Quijano FA, Martínez-Vega RA. Experiencia Y Percepción del Riesgo Asociados a conocimientos, actitudes y Prácticas sobre dengue en Riohacha, Colombia. Ciência & Saúde Coletiva. 2020; 25(3):1137–46. https://doi.org/10.1590/1413-81232020253.08592018 PMID: 32159681
- Ahmad Zamzuri M 'Ammar, Abd Majid FN, Dapari, Hassan MR, Isa AM. Perceived risk for dengue infection mediates the relationship between attitude and practice for dengue prevention: A study in Seremban, Malaysia. International Journal of Environmental Research and Public Health. 2022; 19 (20):13252. https://doi.org/10.3390/ijerph192013252 PMID: 36293856
- Herbuela VR, de Guzman FS, Sobrepeña GD, Claudio AB, Tomas AC, Arriola-delos Reyes CM, et al. Knowledge, attitude, and practices regarding dengue fever among pediatric and adult in-patients in Metro Manila, Philippines. International Journal of Environmental Research and Public Health. 2019; 16 (23):4705. https://doi.org/10.3390/ijerph16234705 PMID: 31779171
- 59. Hairi F, Ong C-HS, Suhaimi A, Tsung T-W, bin Anis Ahmad MA, Sundaraj C, et al. A knowledge, attitude and practices (KAP) study on dengue among selected rural communities in the kuala kangsar district. Asia Pacific Journal of Public Health. 2003; 15(1):37–43. <u>https://doi.org/10.1177/101053950301500107</u> PMID: 14620496
- Zaki R, Roffeei SN, Hii YL, Yahya A, Appannan M, Said MA, et al. Public perception and attitude towards dengue prevention activity and response to dengue early warning in Malaysia. PLOS ONE. 2019; 14(2). https://doi.org/10.1371/journal.pone.0212497 PMID: 30818394
- Caprara A, Lima JW, Marinho AC, Calvasina PG, Landim LP, Sommerfeld J. Irregular water supply, household usage and dengue: A bio-social study in the Brazilian Northeast. Cadernos de Saúde Pública. 2009; 25(suppl 1). https://doi.org/10.1590/s0102-311x2009001300012 PMID: 19287857
- 62. Oliveira KK, Caprara A. Face social do controle do aedes: Em um bairro periférico de fortaleza, Brasil, as Mulheres Tomam a Palavra. Ciência & Saúde Coletiva. 2019; 24(8):2983–92. <u>https://doi.org/10.1590/1413-81232018248.21522017 PMID</u>: 31389545
- 63. Kaiser AJ. "It's complete chaos": Brazilian state overwhelmed by rash of gang violence. The Guardian [Internet]. 2019 Jan 9 [cited 2023 Apr 19]; Available from: https://www.theguardian.com/world/2019/jan/09/brazil-ceara-violence-fortaleza-gangs-bolsonaro.
- 64. Bancroft D, Power GM, Jones RT, Massad E, Iriart JB, Preet R, et al. Vector Control Strategies in Brazil: A qualitative investigation into community knowledge, attitudes and perceptions following the 2015– 2016 zika virus epidemic. BMJ Open. 2022; 12(1). https://doi.org/10.1136/bmjopen-2021-050991 PMID: 35105618
- Touchton M, Wampler B. Democratizing Public Health: Participatory policymaking institutions, Mosquito Control, and Zika in the Americas. Tropical Medicine and Infectious Disease. 2023; 8(1):38. <u>https://doi.org/10.3390/tropicalmed8010038 PMID: 36668945</u>
- 66. Rosenbaum J, Gayle C, Lloyd LS, Rawlins S, Chadee DD, Nathan MB, et al. Community participation in dengue prevention and control: A survey of knowledge, attitudes, and practice in Trinidad and Tobago. The American Journal of Tropical Medicine and Hygiene. 1995; 53(2):111–7. https://doi.org/10.4269/ajtmh.1995.53.111 PMID: 7677210
- 67. Aerts C, Revilla M, Duval L, Paaijmans K, Chandrabose J, Cox H, et al. Understanding the role of disease knowledge and risk perception in shaping preventive behavior for selected vector-borne diseases in Guyana. PLOS Neglected Tropical Diseases. 2020; 14(4). https://doi.org/10.1371/journal.pntd. 0008149 PMID: 32251455
- Elsinga J, Schmidt M, Lizarazo EF, Vincenti-Gonzalez MF, Velasco-Salas ZI, Arias L, et al. Knowledge, attitudes, and preventive practices regarding dengue in Maracay, Venezuela. The American Journal of

Tropical Medicine and Hygiene. 2018; 99(1):195–203. <u>https://doi.org/10.4269/ajtmh.17-0528</u> PMID: 29848406

- 69. Castro M, Sánchez L, Pérez D, Sebrango C, Shkedy Z, Van der Stuyft P. The relationship between economic status, knowledge on Dengue, risk perceptions and practices. PLoS ONE. 2013; 8(12). <u>https://doi.org/10.1371/journal.pone.0081875 PMID: 24349145</u>
- 70. Saied KG, Al-Taiar A, Altaire A, Alqadsi A, Alariqi EF, Hassaan M. Knowledge, attitude and preventive practices regarding dengue fever in rural areas of Yemen. International Health. 2015; 7(6):420–5. https://doi.org/10.1093/inthealth/ihv021 PMID: 25858280
- 71. Daag JV, Ylade M, Jadi R, Adams C, Cuachin AM, Alpay R, et al. Performance of dried blood spots compared with serum samples for measuring dengue seroprevalence in a cohort of children in Cebu, Philippines. The American Journal of Tropical Medicine and Hygiene. 2021; 104(1):130–5. <u>https://doi.org/10.4269/ajtmh.20-0937 PMID</u>: 33146119
- 72. Krentel A, Basker N, Beau de Rochars M, Bogus J, Dilliott D, Direny AN, et al. A multicenter, community-based, mixed methods assessment of the acceptability of a triple drug regimen for elimination of lymphatic filariasis. PLOS Neglected Tropical Diseases. 2021; 15(3). https://doi.org/10.1371/journal. pntd.0009002 PMID: 33657090